Stratified Belief Bases Revision with Argumentative Inference

Marcelo Alejandro Falappa, Alejandro Javier García, Gabriele Kern-Isberner & Guillermo Ricardo Simari

Journal of Philosophical Logic

Published under the Auspices of the Association for Symbolic Logic

ISSN 0022-3611 Volume 42 Number 1

J Philos Logic (2013) 42:161-193 DOI 10.1007/s10992-011-9217-z JOURNAL OF
PHILOSOPHICAL LOGIC

EDITORS
HANS VAN DITMARSCH
BRANDEN FITELSON
JOHN HORTY
and GREG RESTALL



Your article is protected by copyright and all rights are held exclusively by Springer Science+Business Media B.V.. This e-offprint is for personal use only and shall not be self-archived in electronic repositories. If you wish to self-archive your work, please use the accepted author's version for posting to your own website or your institution's repository. You may further deposit the accepted author's version on a funder's repository at a funder's request, provided it is not made publicly available until 12 months after publication.



Stratified Belief Bases Revision with **Argumentative Inference**

Marcelo Alejandro Falappa · Alejandro Javier García · Gabriele Kern-Isberner · Guillermo Ricardo Simari

Received: 27 June 2011 / Accepted: 2 November 2011 / Published online: 25 November 2011 © Springer Science+Business Media B.V. 2011

Abstract We propose a revision operator on a stratified belief base, i.e., a belief base that stores beliefs in different strata corresponding to the value an agent assigns to these beliefs. Furthermore, the operator will be defined as to perform the revision in such a way that information is never lost upon revision but stored in a stratum or layer containing information perceived as having a lower value. In this manner, if the revision of one layer leads to the rejection of some information to maintain consistency, instead of being withdrawn it will be kept and introduced in a different layer with lower value. Throughout this development we will follow the principle of minimal change, being one of the important principles proposed in belief change theory, particularly emphasized in the AGM model. Regarding the reasoning part from the stratified belief base, the agent will obtain the inferences using an argumentative formalism.

M. A. Falappa (🖾) · A. J. García · G. R. Simari Artificial Intelligence Research and Development Laboratory, Department of Computer Science and Engineering, Universidad Nacional del Sur, (8000) Bahia Blanca, Buenos Aires, Argentina e-mail: mfalappa@cs.uns.edu.ar

A. J. García e-mail: ajg@cs.uns.edu.ar

G. R. Simari e-mail: grs@cs.uns.edu.ar

M. A. Falappa · A. J. García Consejo Nacional de Investigaciones Científicas y Técnicas (CONICET), Buenos Aires, Argentina

G. Kern-Isberner Technische Universitaet Dortmund, Department of Computer Science, University of Dortmund, 44221 Dortmund, Germany e-mail: gabriele.kern-isberner@cs.uni-dortmund.de



Thus, the argumentation framework will decide which information prevails when sentences of different layers are used for entailing conflicting beliefs. We will also illustrate how inferences are changed and how the status of arguments can be modified after a revision process.

Keywords Belief revision · AGM model · Argumentation systems

1 Introduction

Belief Revision is the process of changing beliefs by taking into account a new piece of information. The formalization of Belief Revision has been the subject of investigation in Philosophy and Logic, spreading to Computer Science where the research takes place in the areas of Artificial Intelligence and Databases. Argumentation is concerned primarily with accepting claims supported by arguments that are obtained by reasoning from accepted premises; acceptance proceeds in the context of disagreement where the support and interference of other arguments is analyzed. In settings where autonomous agents are involved, Belief Revision describes the way in which an agent is supposed to change its beliefs when new information arrives or when changes in the world are observed restoring the consistency of the knowledge base; Argumentation is concerned with establishing the agent's current beliefs from a potentially incomplete and/or inconsistent knowledge base maintaining the consistency of these beliefs.

It is tempting to take the standard AGM approach [1] for a direct comparison of Argumentation and Belief Revision. This approach, based on partial meet contractions, and all its variants: safe contractions [2], epistemic entrenchment [35], sphere systems [38], and kernel contractions [40], have been the most influential works in belief change and the dynamics of knowledge; no other work has influenced the development of belief revision theory in a similar way. The AGM postulates [1, 33] offer a clear framework that makes fundamental views explicit. The use of beliefs sets, i.e., deductively closed sets of formulas, is one of the basic elements of the standard AGM theory. Beliefs sets, however, are not useful as a formal device for developing an argumentation theory since the basic reasoning steps are abstracted away; this eliminates the fundamental distinction between explicit and implicit beliefs. Precisely, establishing which are the implicit beliefs represents the main focus of an argumentation theory. Thus, the elements contained on the formalisms used in belief base revision [27, 41], or in iterated epistemic change [18, 19, 42, 43] offer a richer and more appropriate base for our research, allowing us to obtain deeper insights into the change processes at all levels.

The connection between belief revision and argumentation was analyzed by Doyle in his seminal work on reason maintenance [22]. Later, his analysis was expanded in [23] where their similarities and differences were thoroughly discussed. In this work, Doyle contrasted the foundations approach and the coherence approach. The first one exemplified in the paper by his own *reason*



maintenance system (also known as *Truth Maintenance System* or TMS), and the latter represented by the AGM theory. He concludes listing these similarities and remarking that: "Perhaps the most fruitful way of viewing the issue is to focus on the great and fundamental similarities of the approaches rather than on the apparently minor differences." While the philosophical discussion about these two forms of considering the representation and revision of beliefs is outside the scope of this paper, it is interesting to note that both perspectives have positively influenced each other [34].

In another line of work of the effort seeking to show connections between systems, the interrelation between belief revision and nonmonotonic inference was made clear in [47], and was further developed for iterative change operations in [44]; this is the BRDI¹ approach to belief revision [24]. As to argumentation, those approaches that are based on rules or use some sort of inference mechanism provide logical chains that establish links between what is accepted as premises and what is concluded. Regarding this aspect, belief revision theories remain on an abstract level, describing by postulates what good inferences are, while argumentation is more concerned with how conclusions are drawn, making explicit the reasons for belief.

One of the first approaches to connect belief revision and argumentation was presented in [29]. The knowledge representation language introduced recognizes *undefeasible* beliefs, represented as K, and *defeasible* beliefs, represented as Δ . In that framework, the reuse of rejected undefeasible conditionals from K is proposed by changing the type of relation between its components. Thus, an undefeasible conditional $\alpha \to \beta$ representing a strict relation between its antecedent and its consequent, instead of being eliminated from K during a change was transformed in a *defeasible conditional* $\alpha \leadsto \beta$ and stored in the set Δ of defeasible beliefs.

A stratified belief base is a belief base in which all beliefs are assigned a certain value and each stratum contains all the beliefs that share the same appraisal. The concept of value will remain as an abstract notion in this work providing a natural way to label each stratum; the set of labels will be assumed to have a total order. Furthermore, and characterizing our approach, each stratum is required to be internally consistent. Notice that direct inconsistency between information stored in different strata could remain. Inconsistency among potential beliefs will be handled by the reasoning mechanism we will introduce leading to a consistent set of actual beliefs.

Such a stratified belief base provides a suitable foundation for the epistemic state of an intelligent agent. This epistemic state will contain explicit beliefs, i.e., the ones stored in the stratified belief base, and implicit beliefs, i.e., the ones that can be obtained by inference from the explicit beliefs. For the derivation of the implicit beliefs from such bases, we will resort to an

²For a different, paraconsistent approach see [7–9], see Section 6 for brief datails.



¹BRDI: Belief Revision as Defeasible Inference.

argumentation system approach under one of Dung's style semantics [25]. Since each stratum is assumed to be consistent, it is necessary to devise a mechanism for revising stratified belief bases for handling the case when conflicting information attempts to be added to the same stratum. The decision to keep each stratum consistent, makes it necessary for a rational agent to resolve these conflicts before obtaining inferences from its stratified epistemic base. Our intention is to generalize the one-level reuse of beliefs shown in [29] by extending it to a multi-level formalism, i.e., beliefs that are discarded from one stratum during revision will provide the input for the revision of the next, less valued stratum. In this manner, beliefs are not given up completely, but can be preserved in a weakened form. We will exemplify how this reuse is relevant for the derivation of beliefs, and how argumentative inferences are modified after changes.

This paper joins the research effort aimed to understand the connections between Belief Revision and Argumentation by offering the following contributions: (i) the definition of a revision operator on stratified belief bases and several postulates characterizing it, (ii) the multi-level reuse of discarded beliefs from the revision of each stratum and, as a result of this, (iii) the ability to dynamically change the level of beliefs, (iv) the integration of an argumentation formalism that allows for warranted inferences taking into consideration pieces of information with different values.

The presentation is organized as follows. Section 2 provides opening material and basic notation on stratified belief bases. Section 3 shows the argumentation framework that will be used for reasoning. Section 4 presents a revision subsystem that will be used to preserve consistency on every stratum of a stratified belief base. Section 5 introduces a revision operator to be applied to a stratified belief base and presents some postulates for this operator addressing connections between belief revision and argumentation, Section 6 discusses pertinent related work, and Section 7 offers the conclusions of this paper.

2 Stratified Belief Bases: Preliminaries and Notation

In classical belief revision models, some sentences that are inconsistent with the epistemic input are fully discarded. This decision seems to be going against the idea that is good to preserve as much old information as possible. This criterion is known as the *principle of minimal change* [33] in the related literature. The multi-level reuse of beliefs offers the opportunity of taking those sentences that are eliminated by the revision process in one stratum and using them to revise the next less valued stratum. This reuse mechanism avoids the complete loss of that information, offering the advantage of having a *dynamic classification of beliefs*, that is, beliefs are dynamically classified using their value. The process of dynamic classification has been frequently used in the evolution of human knowledge. For instance, the belief establishing that all metals are solid under normal conditions of pressure and temperature was maximally reliable, i.e., having the highest value, for centuries. However, at



some point in history, it was discovered that mercury is a metal that is in liquid state under afore said conditions, and the discovery effected a change in the belief describing that property of metals. Many other beliefs have changed in similar manner throughout the evolution of Science and undoubtedly will continue happening.

In this paper we will adopt a propositional language \mathcal{L} with a complete set of boolean connectives: \neg , \wedge , \vee , \rightarrow , \leftrightarrow . Formulæ in \mathcal{L} will be denoted by lowercase Greek letters: α , β , δ , ..., ω . Sets of sentences in \mathcal{L} will be denoted by uppercase Latin letters: A, B, C, ..., Z. The symbol \top represents a tautology or truth, and the symbol \bot represents a contradiction or falsum. The characters γ and σ will be reserved to represent selection and incision functions for change operators respectively. We use a consequence operator $Cn: 2^{\mathcal{L}} \longrightarrow 2^{\mathcal{L}}$ that takes sets of sentences in \mathcal{L} and produces new sets of sentences in \mathcal{L} . This operator satisfies inclusion ($A \subseteq Cn(A)$), idempotence (Cn(A) = Cn(Cn(A))), and monotony (if $A \subseteq B$ then $Cn(A) \subseteq Cn(B)$). It is assumed that the consequence operator includes classical consequences and verifies the standard properties of supraclassicality (if α can be derived from A by deduction in classical logic, then $\alpha \in Cn(A)$), deduction ($\beta \in Cn(A \cup \{\alpha\})$) if and only if $(\alpha \to \beta) \in Cn(A)$) and compactness (if $\alpha \in Cn(A)$ then $\alpha \in Cn(A')$ for some finite subset A' of A). In general, we will write $\alpha \in Cn(A)$ as $A \vdash \alpha$.

In our approach, an agent's epistemic state will be represented as a stratified belief base. In such base, each stratum will contain a set of sentences with the same value and each stratum will be assumed consistent. This decision contrast with other approaches, such as [7,8] where a form of argumentation is used as a general inference mechanism from a (potentially) inconsistent stratified belief base rendering a paraconsistent approach.

Definition 1 $\Sigma = (\Sigma_0, ..., \Sigma_n)$ is a *stratified belief base* (SBB) if and only if $\Sigma_i \subseteq \mathcal{L}$ and Σ_i is finite and consistent for all $0 \le i \le n$, and all the sentences in Σ_i have the same value. The strata are considered to be ordered in the following way: the beliefs of Σ_j have a higher assigned value than the ones in Σ_i when j < i. Sometimes, slightly abusing the notation, $x \in \Sigma$ will mean that $x \in \Sigma_i$ for some $\Sigma_i \in \Sigma$, $0 \le i \le n$.

It seems a good design decision for a rational agent, faced with contradictory beliefs of different levels, should opt to believe only the stronger of them disregarding the weaker. Such agent will reserve the first stratum Σ_0 for proper beliefs, while the rest of the layers refer to information that: (i) it has assigned a lower level than the information in Σ_0 , (ii) it is believed only when it does not contradict beliefs that are in Σ_0 or that are contained in a strata of higher level, and (iii) even when it is believed, it will remain more provisional than any belief in Σ_0 . Therefore, in our approach, stratified belief bases are largely composed of information that is not fully believed, and the stratification is produced by the perceived value of that information. This characterization differs from other approaches that consider that beliefs have a certain strength or entrenchment [35, 46, 57].



It is important to note that our approach differs from the research presented in [7, 8]. In these works, our abstract notion of value is grounded as reliability and each stratum can be inconsistent. Moreover, the reliability assigned to the elements in the belief base cannot be modified. Our approach considers consistent strata and will allow to change the level of beliefs as a consequence of a revision.

3 Warranted Inference Through Argumentation

Stratified belief bases (SBB) are collections of sentences with different values attached, possibly containing multiple copies of sentences in different strata, each stratum being consistent, and the union of the set of strata could be inconsistent, i.e., sentences in one stratum can contradict sentences in a different stratum. To make good use of this information in selecting the implicit beliefs, a mechanism with the ability of prioritizing the beliefs with higher value and with the capability of solving conflicts is needed; argumentation frameworks were developed for that purpose. In this work, we introduce a formalism that defines an argumentation framework that allows to entail warranted conclusions from a stratified belief base.

The use of argumentation will introduce the possibility of pondering arguments for and against a given conclusion and extending the analysis to interference occurring in the intermediate steps of the reasoning process, i.e., at the level of sub-arguments. After presenting the argumentation framework, in Section 5 a revision operator for a stratified belief base will be introduced. This operator will allow the addition of incoming information to the SBB, maintaining the local consistency of each SBB's stratum and handling the change in the value assigned of each sentence.

3.1 Arguments from Stratified Belief Bases

Arguments are pieces of reasoning that support conclusions from evidence. They can have intermediate reasoning steps, and hence, can be structured with sub-arguments. Each sub-argument provides support for an intermediate conclusion. In the literature, structured arguments are assumed to be consistent, in the sense that an argument does not contain two sub-arguments that support contradictory conclusions. In other words, the set of formulas used in the reasoning to infer the conclusion is assumed to be consistent [10, 11, 52, 58]. Thus, an argument cannot be in conflict with itself, i.e., cannot be self-defeating, providing a consistent explanation for the conclusion it supports.

Our definition of an argument is adapted from the notion of argument structure introduced in [58] and widely accepted in many formalizations that consider how arguments are built [6, 55]. The intuitions behind these definitions are simple. An argument for a conclusion α is a minimal set of consistent formulas from the stratified belief base Σ that entails the conclusion. Furthermore, pondering the entity that builds the argument, lets consider an



agent whose beliefs are represented as Σ . For this agent, Σ_0 contains its more valued beliefs, therefore it is assumed that this stratum represents a set of indisputable beliefs; they provide the foundation for the other accepted beliefs. Following the same conceptualization, no argument should contradict the sentences in Σ_0 . The definitions and remarks below formalize the above mentioned intuitions.

Definition 2 Let $\Sigma = (\Sigma_0, ..., \Sigma_n)$ be a stratified belief base and α a sentence. A set of sentences A is an *argument* for α from Σ , denoted $\langle A, \alpha \rangle_{\Sigma}$ or simply $\langle A, \alpha \rangle$, if:

- 1. $A \subseteq (\Sigma_0 \cup \ldots \cup \Sigma_n)$.
- 2. $A \cup \Sigma_0$ is consistent.
- 3. $\alpha \in Cn(A \cup \Sigma_0)$.
- 4. There is no $X \subset A$ that satisfies conditions 2 and 3.

Given the argument $\langle A, \alpha \rangle_{\Sigma}$, the sentence α is called the *conclusion* of the argument, and the set A is called the *support* of α . Sometimes, when no confusion could arise, we will simplify the notation for arguments as $\langle A, \alpha \rangle$.

Definition 3 Let Σ be a stratified belief base. An argument $\langle B, \beta \rangle_{\Sigma}$ is a *sub-argument* of $\langle A, \alpha \rangle_{\Sigma}$, if $B \subseteq A$.

Observation 1 Given an argument $\langle A, \alpha \rangle_{\Sigma}$ built from a stratified belief base $\Sigma = (\Sigma_0, \dots, \Sigma_n)$, the condition 2 of Definition 2 determines that $\neg \alpha \notin Cn(\Sigma_0)$.

Observation 2 Let Σ be a stratified belief base, and let $\langle A, \alpha \rangle_{\Sigma}$ be an argument from Σ with sub-arguments $\langle B, \beta \rangle_{\Sigma}$, and $\langle C, \gamma \rangle_{\Sigma}$. Since A is consistent (Definition 2), $B \cup C$ is a consistent set of formulas.

Example 1 Consider the stratified belief base $\Sigma = (\Sigma_0, \Sigma_1, \Sigma_2, \Sigma_3)$, where:

$$\Sigma_{0} = \begin{cases} k \\ h \\ h \to j \end{cases} \qquad \Sigma_{1} = \begin{cases} b \to c \\ e \to \neg b \\ h \to b \\ k \to \neg f \end{cases} \qquad \Sigma_{2} = \begin{cases} e \\ a \to b \\ f \to \neg c \\ h \to f \end{cases} \qquad \Sigma_{3} = \begin{cases} a \\ j \end{cases}$$

The following arguments are only some of the arguments that can be obtained from Σ :

$$\begin{array}{lll} \langle \{k\},k\rangle & & \langle \{A_3,\neg b\rangle & A_3=\{e,e\rightarrow\neg b\} \\ \langle \{e\},e\rangle & & \langle A_4,\neg c\rangle & A_4=\{h,h\rightarrow f,f\rightarrow\neg c\} \\ \langle A_1,c\rangle & A_1=\{a,a\rightarrow b,b\rightarrow c\} & \langle A_5,f\rangle & A_5=\{h,h\rightarrow f\} \\ \langle A_2,b\rangle & A_2=\{a,a\rightarrow b\} & \langle A_7,\neg f\rangle & A_7=\{k,k\rightarrow\neg f\} \end{array}$$

As an example, note that $\langle A_2, b \rangle$ is a sub-argument of both $\langle A_1, c \rangle$ and $\langle A_0, \neg e \rangle$, and that $\langle A_5, f \rangle$ is a sub-argument of $\langle A_4, \neg c \rangle$.



Since arguments can contain sentences of different strata, the comparison of arguments will take account of such information. The value of a set of sentences will depend on the strata the sentences belong to. Arguments will have a higher value if they are built with sentences of a stratum with a lower index. Since a sentence can belong to more of one stratum, the stratum with lower index (greatest value) should be chosen. Hence, the *index* of a sentence β in Σ , denoted $index_{\Sigma}(\beta)$, will be the index corresponding to the lowest indexed stratum – and hence most valued – in Σ to which β belongs to.

Definition 4 Let $\Sigma = (\Sigma_0, ..., \Sigma_n)$ be a stratified belief base such that $\beta \in (\Sigma_0 \cup ... \cup \Sigma_n)$. Then, $index_{\Sigma}(\beta) = i$ if and only if $\beta \in \Sigma_i$ and there is no Σ_i (i < i) such that $\beta \in \Sigma_i$. We say that i is the *stratum* of β .

The stratum of an argument is therefore defined by the index of its weakest sentence.

Definition 5 Let Σ be a stratified belief base and $A = \{x_1, x_2, ..., x_m\}$ be a set of sentences in Σ . The *stratum of the set A* is

$$str_{\Sigma}(A) = \max\{index_{\Sigma}(x_1), index_{\Sigma}(x_2), \dots, index_{\Sigma}(x_m)\}\$$

Example 2 Consider the Σ in Example 1, then:

```
\begin{array}{ll} index_{\Sigma}(k)=0, & index_{\Sigma}(j)=3, & index_{\Sigma}(h\rightarrow b)=1,\\ str_{\Sigma}(\langle\{k\},k\rangle)=0, & str_{\Sigma}(\langle\{h\},h\rangle)=0, & str_{\Sigma}(\langle\{e\},e\rangle)=2,\\ str_{\Sigma}(\langle\{j\},j\rangle)=3, & str_{\Sigma}(\langle\{h,h\rightarrow j\},j\rangle)=0, & str_{\Sigma}(\langle A_1,c\rangle)=3,\\ str_{\Sigma}(\langle A_2,b\rangle)=3, & str_{\Sigma}(\langle A_3,\neg b\rangle)=2, & str_{\Sigma}(\langle A_4,\neg c\rangle)=2,\\ str_{\Sigma}(\langle A_5,f\rangle)=2, & str_{\Sigma}(\langle A_6,b\rangle)=1, & str_{\Sigma}(\langle A_7,\neg f\rangle)=1 \end{array}
```

Next, based on $str_{\Sigma}(\cdot)$, we introduce the order " \succeq_{Σ} " in the set of arguments that can be built from Σ . Recall that a lower number represents more value.

Definition 6 Let $\langle B, \beta \rangle_{\Sigma}$ and $\langle A, \alpha \rangle_{\Sigma}$ be two arguments from a stratified belief base Σ . We will say that $\langle B, \beta \rangle_{\Sigma}$ is as least as good than $\langle A, \alpha \rangle_{\Sigma}$, denoted as $\langle B, \beta \rangle_{\Sigma} \succeq_{\Sigma} \langle A, \alpha \rangle_{\Sigma}$, if and only if $str_{\Sigma}(B) \leq str_{\Sigma}(A)$.

The following proposition establishes that a sub-argument cannot be weaker than any of its super-arguments.

Proposition 1 Let Σ be a stratified belief base, if $\langle B, \beta \rangle_{\Sigma}$ is a sub-argument of $\langle A, \alpha \rangle_{\Sigma}$ then $\langle B, \beta \rangle_{\Sigma} \succeq_{\Sigma} \langle A, \alpha \rangle_{\Sigma}$.

Proof Suppose that $\langle B, \beta \rangle_{\Sigma} \not\succeq_{\Sigma} \langle A, \alpha \rangle_{\Sigma}$, then it should be the case that $str_{\Sigma}(B) > str_{\Sigma}(A)$. However, if $str_{\Sigma}(B) > str_{\Sigma}(A)$, then there should be some sentence in B that is not in A, but that is not possible because $B \subseteq A$.



3.2 Conflict and Defeat Relations

Two arguments that support contradictory conclusions clearly interfere with each other; this will be considered as a relation of *conflict*. For instance, the arguments $\langle A_2, b \rangle$ and $\langle A_3, \neg b \rangle$ from Example 1 are in conflict. Furthermore, given two arguments A and B that are in conflict, such as $B \succeq_{\Sigma} A$, it is natural to establish that the interference of B over A is successful; this introduces the notion of *defeat*.

Defeat can be effected in different ways. In presenting the idea, we have introduced a particular type of defeat where the attack is aimed at the conclusion, this form of defeat is called *rebuttal*; other kinds of defeat exists leading to a richer relation. When the attack is indirect, i.e., being aimed at an intermediate step in the reasoning, the defeat is called *undercut*. In this work, undercuts will be accomplished by the attack of one argument to a subargument of another argument, that is, it is a rebuttal over a sub-argument of the attacked argument. There is another form of attack, called *assumption attack*. Given that we will not use assumptions in our formalism, this latter form of attack will not appear in our presentation. The formal definitions of conflict and defeat are introduced next.

Definition 7 Let Σ be a stratified belief base. Two arguments $\langle A, \alpha \rangle_{\Sigma}$ and $\langle B, \beta \rangle_{\Sigma}$ are *in conflict*, if $\Sigma_0 \cup \{\alpha, \beta\}$ is an inconsistent set.

We would like to discuss the particular use of the stratum Σ_0 as context to analyze the inconsistency of the conclusions of two arguments. The Σ_0 stratum is a set of consistent beliefs with maximum value and any argument, by its construction, should be consistent with it. Since each conclusion by itself is consistent with Σ_0 , the conclusions must be contradictory with each other, possibly through the use of an appropriated subset of the beliefs in Σ_0 . The example below shows the general case.

Example 3 Consider the stratified belief base $\Sigma = (\Sigma_0, \Sigma_1)$, where:

$$\Sigma_0 = \begin{cases} x \\ y \\ a \to c \\ b \to \neg c \end{cases} \qquad \Sigma_1 = \begin{cases} x \to a \\ y \to b \end{cases}$$

The arguments $\langle A, a \rangle$ with $A = \{x, x \rightarrow a\}$ and $\langle B, b \rangle$ with $B = \{y, y \rightarrow b\}$ are in conflict even though their conclusions are not in direct contradiction.

Definition 8 Let Σ be a stratified belief base. An argument $\langle B, \beta \rangle_{\Sigma}$ defeats $\langle A, \alpha \rangle_{\Sigma}$, if there exists a sub-argument $\langle C, \gamma \rangle_{\Sigma}$ of $\langle A, \alpha \rangle_{\Sigma}$, such that the following two conditions hold:

- $-\langle B, \beta \rangle_{\Sigma}$ and $\langle C, \gamma \rangle_{\Sigma}$ are in conflict, and
- $\langle B, \beta \rangle_{\Sigma} \succeq_{\Sigma} \langle C, \gamma \rangle_{\Sigma}.$

We will say that $\langle B, \beta \rangle_{\Sigma}$ defeats $\langle A, \alpha \rangle_{\Sigma}$ at sub-argument $\langle C, \gamma \rangle_{\Sigma}$.



Example 4 Consider the stratified belief base of Example 1. Argument $\langle A_3, \neg b \rangle$ defeats $\langle A_1, c \rangle$ (at the sub-argument $\langle A_2, b \rangle$), and $\langle A_6, b \rangle$ defeats $\langle A_3, \neg b \rangle$. Argument $\langle A_4, \neg c \rangle$ defeats $\langle A_1, c \rangle$ and $\langle A_7, \neg f \rangle$ defeats $\langle A_4, \neg c \rangle$ (at the sub-argument $\langle A_5, f \rangle$).

Consider two arguments $\langle C, \gamma \rangle_{\Sigma}$ and $\langle A, \alpha \rangle_{\Sigma}$ that are in conflict. Suppose that $str_{\Sigma}(C) = str_{\Sigma}(A)$. Then it holds that $\langle C, \gamma \rangle_{\Sigma} \succeq_{\Sigma} \langle A, \alpha \rangle_{\Sigma}$ and also that $\langle A, \alpha \rangle_{\Sigma} \succeq_{\Sigma} \langle C, \gamma \rangle_{\Sigma}$. Hence, in this particular case, it holds that $\langle C, \gamma \rangle_{\Sigma}$ defeats $\langle A, \alpha \rangle_{\Sigma}$ and $\langle A, \alpha \rangle_{\Sigma}$ defeats $\langle C, \gamma \rangle_{\Sigma}$.

As a consequence of the way the comparison criterion " \succeq_{Σ} " is defined, if B defeats A, then B will also defeat every super-argument of A, that is, B also defeats every argument C such that A is a sub-argument of C (see Proposition 1).

Observation 3 Let Σ be a stratified belief base. If $\langle B, \beta \rangle_{\Sigma}$ is a sub-argument of $\langle A, \alpha \rangle_{\Sigma}$ and $\langle C, \gamma \rangle_{\Sigma}$ is a sub-argument of $\langle A, \alpha \rangle_{\Sigma}$, then $\langle B, \beta \rangle_{\Sigma}$ cannot defeat $\langle C, \gamma \rangle_{\Sigma}$ and vice versa. (see Observation 2).

Proposition 2 Let $\langle A, \alpha \rangle_{\Sigma}$ be an argument. If $A \subseteq \Sigma_0$ then there is no argument $\langle B, \beta \rangle$ such that $\langle B, \beta \rangle_{\Sigma}$ defeats $\langle A, \alpha \rangle_{\Sigma}$.

Proof An argument B can only defeat A if there exists a sub-argument C of A such that B is not weaker than C. Since $A \subseteq \Sigma_0$, $str_{\Sigma}(A) = 0$, and any sub-argument C will be such that $str_{\Sigma}(C) = 0$. Suppose that there exists an argument B that is a defeater of A at sub-argument C, then B is not weaker than C, i.e., $str_{\Sigma}(B) = str_{\Sigma}(C) = 0$, but that means that $B \subseteq \Sigma_0$. But B is a defeater of A at C, that is, the conclusions of B and C are contradictory. Since Σ_0 is consistent the existence of B is not possible.

Corollary 1 Given an argument $\langle A, \alpha \rangle_{\Sigma}$. If $str_{\Sigma}(A) = 0$ then there exists no argument that can defeat $\langle A, \alpha \rangle_{\Sigma}$.

Proof It suffices to observe that if $str_{\Sigma}(A) = 0$ then $A \subseteq \Sigma_0$, and in that case, by Proposition 2, argument A can have no defeater.

3.3 Argumentative Inference

The argumentation formalism introduced above allows us to build the set of arguments that will be involved in the process of deciding what the implicit beliefs are. In argumentation terminology this is the set of beliefs supported by warranted arguments. For obtaining these beliefs, we will make use of the semantics defined for abstract argumentation frameworks.

An argumentation framework [25] is a pair $(\mathfrak{Args}, \mathbf{R})$, where \mathfrak{Args} is a finite set of arguments and \mathbf{R} is a binary relation between arguments such that $\mathbf{R} \subseteq \mathfrak{Args} \times \mathfrak{Args}$. The notation $(A, B) \in \mathbf{R}$ (or, equivalently, $A \mathbf{R} B$) means that A attacks B. It is interesting to note that the Dung's relation called attack in his



original paper corresponds to the notion of defeat introduced in this work, i.e., in Dung's formalism every attack was successful, therefore we will consistently use defeat instead of attack.

An argumentation semantics is the formal definition of the argument evaluation process leading to decide which arguments are able to survive the attacks defined in the framework. These "survivors" will be considered as being able to support their conclusions. The research has produced two different ways of carrying out the evaluation, namely extension-based and labeling-based argumentation semantics. The first one provides a declarative definition, meanwhile the second one is procedural in nature. An extension is a subset of arguments contained in the framework, and the extension-based approach specifies how to obtain the subsets that form the set of extensions. Every extension contains a set of arguments that together can be acceptable in the context of the attack relation. The labeling-based approach provides a way of assigning a label to each argument in the framework choosing that label from an appropriated set, such as {in, out, undecided}. The assignment of labels yields a set of labelings that correspond to the extensions found through the declarative method.

Dung [25] introduces several argumentation semantics that provide a way of evaluating the status of the arguments in the framework constructing extensions, e.g., complete, grounded, stable, and preferred semantics. Other semantics have been proposed after the initial definition, e.g., stage, semistable, ideal, *CF*2, and prudent semantics. For further details on recent developments see [4].

Given a stratified belief base Σ and using the argumentation formalism defined above, an argumentation framework $(\mathfrak{Args}, \mathbf{R})$ can be obtained, and some skeptical semantics can be applied in order to obtain warranted conclusions. An argument will be skeptically warranted if and only if it is warranted in all the extensions produced by the chosen semantics. Grounded semantics is an example of skeptical semantics, and we will use it in the example below.

Example 5 Consider again Example 1. The argumentation framework $(\mathfrak{Args}, \mathbf{R})$ can be obtained, where $\{A_0, A_1, A_2, A_3, A_4, A_5, A_6, A_7\} \subseteq \mathfrak{Args}$ and \mathbf{R} contains, for instance, the following tuples:

$$\{(A_3, A_2), (A_3, A_1), (A_4, A_1), (A_7, A_4), (A_7, A_5), (A_6, A_3)\}$$

If we choose to use grounded semantics [25], the arguments A_1 , A_2 , A_6 and A_7 are warranted, and so all of b, c, and $\neg f$ can be inferred from our stratified belief base, among others.

Note that the particular semantics chosen for obtaining the warranted arguments is a decision that will affect the behavior of the system. For more details regarding different aspects of skeptical semantics in abstract argumentation frameworks see [3, 5]. The reader may obtain a general perspective on argumentation in general and argumentation semantics in particular in [6], or in [10, 55] where deeper and detailed accounts are presented.



4 Multiple Revision Subsystem

A multiple revision operator is defined as a belief revision operator capable of accepting a set of sentences as epistemic input. The intention is to use this type of operator as part of the global framework we are introducing. To construct a multi-layer revision operator, first we need to discuss certain general ideas concerning revision operators on each layer of a stratified belief base seen in isolation. Although some related literature on the topic of multiple revision exists, such as the proposed in [51] where multiple revision is reduced to classical AGM sentence revision, or the work of [21] where revision and expansion operators of logic programs by answer set semantics were presented, our approach differs from these in that we consider multiple revision in a more general, logically classic environment. Further discussion on related work on this particular subject can be found in [28]. We will proceed by recalling the main ideas originally presented in that paper and introducing a generalization of its capabilities.

4.1 The General Setting

The layers in a stratified belief base Σ can be regarded as separated, consistent belief bases. This will allow us to conceptualize the problem as multiple belief base revision.

Let S be a consistent belief base, A and B be consistent sets of sentences, and * be a binary multiple revision operator that takes a belief base and a set of sentences as inputs. The following postulates are proposed for multiple revision operations:

Inclusion: $S*A \subseteq S \cup A$.

Weak Success: If A is consistent then $A \subseteq S*A$.

Relative Success: $A \subseteq S*A \text{ or } S*A = S.$

Consistency: If A is consistent then S*A is consistent. Vacuity 1: If A is inconsistent then S*A = S. Vacuity 2: If $S \cup A \nvdash \bot$ then $S*A = S \cup A$.

Uniformity 1: Given A and B two consistent sets, for all subset X of S,

if $(X \cup A) \vdash \bot$ if and only if $(X \cup B) \vdash \bot$ then

 $S \setminus (S*A) = S \setminus (S*B).$

Uniformity 2: Given A and B two consistent sets, for all subset X of S,

if $(X \cup A) \vdash \bot$ if and only if $(X \cup B) \vdash \bot$ then

 $S \cap (S*A) = S \cap (S*B)$.

Relevance: If $\alpha \in S \setminus (S*A)$ then there is a set C such that

 $S*A \subseteq C \subseteq (S \cup A)$, C is consistent with A but $C \cup \{\alpha\}$

is inconsistent with A.

Core-Retainment: If $\alpha \in S \setminus (S*A)$ then there is a set C such that

 $C \subseteq (S \cup A)$, C is consistent with A but $C \cup \{\alpha\}$ is incon-

sistent with A.



Inclusion says that any change operator between two arbitrary sets is included in the (unrestricted) union of them. Weak Success gives priority to the new information only when it is consistent. This postulate was inspired by the AGM success postulate with an additional restriction on the input set: it must be consistent. Relative Success establishes that all sentences of the input set are added to the changed set or nothing is changed (except in the limit case in which some sentences of the input set are in the belief base). Consistency determines that the changed set is consistent whenever the input set is consistent. Vacuity 1 says that nothing is changed if the input set is inconsistent. Vacuity 2 establishes that if the input set is consistent with the original belief base, then the changed set is equal to the union of them. Uniformity 1 determines that if two consistent sets are consistent with the same subsets of the original belief base S then the respective erased sentences of S should be identical. Uniformity 2 determines that if two consistent sets are consistent with the same subsets of the original belief base S then the respective preserved sentences of S should be identical. Uniformity 2 is an adaptation of Uniformity [39] which is used in revisions where the epistemic input is just a single sentence. Uniformities 1 and 2 are equivalent and they can be indistinctly used in representation theorems [28]. Relevance and Core-Retainment were adapted from [41] and they express the intuition that nothing is removed from the original belief base unless its removal in some way contributes to making the new belief base consistent. It is easily seen that Relevance is stronger than Core-Retainment, i.e., Relevance implies Core-Retainment. The following lemma shows some relations among postulates [28].

Lemma 1

- (a) If an operator satisfies relevance then it satisfies core-retainment.
- (b) If an operator satisfies vacuity 1 and weak success then it satisfies relative success.
- (c) An operator satisfies uniformity 1 if and only if it satisfies uniformity 2.
- (d) If an operator satisfies inclusion, weak success and core-retainment then it satisfies vacuity 2.

In order to construct the multiple revision operators, we must define the concept of (A-inconsistent) kernels and (A-consistent) remainder sets.

Definition 9 Let S, A be sets of sentences, where A is consistent. The set of A-inconsistent-kernels of S, noted by $S \perp \!\!\! \perp_{\perp} A$, is the set of sets X such that:

- 1. $X \subseteq S$.
- 2. $X \cup A$ is inconsistent.
- 3. For any X', if $X' \subset X \subseteq S$ then $X' \cup A$ is consistent.

That is, given a consistent set A, $S \perp \!\!\! \perp_{\perp} A$ is the set of minimal S-subsets inconsistent with A.



Definition 10 Let S, A be sets of sentences, where A is consistent. The set of A-consistent-remainders of S, noted by $S \perp_{\top} A$, is the set of sets X such that:

- 1. $X \subseteq S$.
- 2. $X \cup A$ is consistent.
- 3. For any X', if $X \subset X' \subseteq S$ then $X' \cup A$ is inconsistent.

That is, $S \perp_{\top} A$ is the set of maximal S-subsets consistent with A.

Example 6 Suppose that $S = \{p, p \to q, q, \neg r\}$ and $A = \{\neg q, r\}$. Then we have that:

- $S \perp \!\!\! \perp_{\perp} A = \{ \{ p, p \to q \}, \{ q \}, \{ \neg r \} \}.$
- $S \perp_{\top} A = \{\{p\}, \{p \to q\}\}.$

4.2 Prioritized Multiple Kernel Revision

The first construction of multiple revision by a set of sentences is based on the concept of A-inconsistent-kernels. In order to complete the construction, we must define an incision function that cuts in every inconsistent-kernel.

Definition 11 Let S be a set of sentences. Then σ is a *consolidated incision* function for $S(\sigma: \mathbf{2}^{2^{\mathcal{L}}} \longrightarrow \mathbf{2}^{\mathcal{L}})$ if and only if, for all consistent set A:

- 1. $\sigma(S \perp \perp A) \subseteq \bigcup S \perp \perp A$.
- 2. If $X \in S \perp \perp_{\perp} A$ then $X \cap (\sigma(S \perp \perp_{\perp} A)) \neq \emptyset$.

Definition 12 Let S, A be sets of sentences, A consistent, and σ a consolidated incision function for S. The *prioritized multiple kernel revision of* S by A that is generated by σ is the operator $*_{\sigma}$ such that $(*_{\sigma}: 2^{\mathcal{L}} \times 2^{\mathcal{L}} \longrightarrow 2^{\mathcal{L}})$:

$$S*_{\sigma} A = \begin{cases} (S \setminus \sigma(S \perp \perp_{\perp} A)) \cup A & \text{if } A \text{ is consistent} \\ S & \text{otherwise} \end{cases}$$

Observation 4 Let S, A be sets of sentences, A be consistent. Suppose that $\alpha \in S$ and $\alpha \in A$. Then $\alpha \notin \bigcup (S \perp \!\!\! \perp_{\perp} A)$ and, therefore, $A \cap \bigcup (S \perp \!\!\! \perp_{\perp} A) = \emptyset$.

From Observation 4 and Definition 11, it follows that all the sentences of A are *protected*, meaning they cannot be considered for removing by the consolidated incision function. That is, a consolidated incision function selects among the sentences of $K \setminus A$ that make $K \cup A$ inconsistent.

Theorem 1 An operator $*_{\sigma}$ is a prioritized multiple kernel revision operator for S if and only if it satisfies inclusion, consistency, weak success, vacuity (1 and 2), uniformity (1 and 2), and core-retainment.



Weak Success ensures the input set is accepted when it is consistent. Vacuity 1 ensures that the belief base remains unchanged when the input set is inconsistent. These two postulates resolve a controversial point of AGM revisions such as success which forces α to be accepted in the revision of S by α even though α might be inconsistent.

4.3 Prioritized Multiple Partial Meet Revision

The second construction of multiple revision by a set of sentences is based on the concept of A-consistent-remainders. In order to complete the construction, we must define a selection function that selects the 'best' consistent remainders.

Definition 13 Let S be a set of sentences. Then γ is a *consolidated selection* function for $S(\gamma: \mathbf{2}^{2^{\mathcal{L}}} \longrightarrow \mathbf{2}^{2^{\mathcal{L}}})$ if and only if, for all set A:

- 1. If $S \perp_{\top} A \neq \emptyset$ then $\gamma(S \perp_{\top} A) \subseteq S \perp_{\top} A$.
- 2. If $S \perp_{\top} A = \emptyset$ then $\gamma(S \perp_{\top} A) = \{S\}$.

Observation 5 Let S, A be sets of sentences, A be consistent. Suppose that $\alpha \in S$ and $\alpha \in A$. Then $\alpha \in X$ for all $X \in S \perp_T A$ and, therefore, $\alpha \in \bigcap (S \perp_T A)$.

From Observation 5 and Definition 13, it follows that all the sentences of $K \cap A$ are *protected*, meaning that they are included in the intersection of any collection of remainders. That is, a consolidated selection function selects a subset of the set $K \perp_{\top} A$ whose elements all contain the set $K \cap A$.

Definition 14 Let S, A be sets of sentences, A consistent, and γ a consolidated selection function for S. The *prioritized multiple partial meet revision of* S by A generated by γ is the operator $*_{\gamma}: \mathbf{2}^{\mathcal{L}} \times \mathbf{2}^{\mathcal{L}} \longrightarrow \mathbf{2}^{\mathcal{L}}$ such that:

$$S*_{\gamma} A = \begin{cases} \bigcap_{\gamma} (S \perp_{\top} A) \cup A \text{ if } A \text{ is consistent} \\ S & \text{otherwise} \end{cases}$$

Theorem 2 An operator $*_{\gamma}$ is a prioritized multiple partial meet revision operator for S if and only if it satisfies inclusion, consistency, weak success, vacuity (1 and 2), uniformity (1 and 2) and relevance.

Corollary 2 Let S be a belief base. If * is a multiple partial meet revision for S then * is a multiple kernel revision for S.

From the definitions, Lemma 1, and Theorems 1 and 2, it is easily seen that multiple partial meet revision operators are multiple kernel revision operators [28].

The following example, adapted from a similar one presented in [41, p. 91], clarifies that the converse of the above corollary does not hold, i.e., that



some multiple kernel revision operators are not multiple partial meet revision operators.

Example 7 Let p, q and r be logically independent sentences and $S = \{p, q, r\}$. Suppose that $A = \{\neg p \lor \neg q, \neg p \lor \neg r\}$. Suppose that we are applying a prioritized multiple kernel revision. Then we have that:

$$S \perp \!\!\! \perp_{\perp} A = \{ \{p, q\}, \{p, r\} \}$$

Suppose that $\sigma(S \perp \!\!\! \perp_{\perp} A) = \{p, r\}$. Then, the prioritized multiple kernel revision of *S* with respect to *A* is:

$$S*_{\sigma} A = (S \setminus \sigma(S \perp \perp \perp A) \cup A = \{q, \neg p \vee \neg q, \neg p \vee \neg r\}$$

Now, suppose that we are applying a prioritized multiple partial meet revision. Then we have that:

$$S \perp_{\top} A = \{\{p\}, \{q, r\}\}$$

If γ is a selection function for S, then $S*_{\gamma}A = \bigcap (S \perp_{\top} A) \cup A$. We have three cases:

$$\gamma_1(S \perp_{\top} A) = \{\{p\}\} \text{ and } S*_{\gamma_1} A = \{p, \neg p \vee \neg q, \neg p \vee \neg r\}.$$
 $\gamma_2(S \perp_{\top} A) = \{\{q, r\}\} \text{ and } S*_{\gamma_2} A = \{q, r, \neg p \vee \neg q, \neg p \vee \neg r\}.$
 $\gamma_3(S \perp_{\top} A) = \{\{p\}, \{q, r\}\} \text{ and } S*_{\gamma_3} A = \{\neg p \vee \neg q, \neg p \vee \neg r\}.$

Thus, we can immediately conclude that $*_{\sigma}$ is not a multiple partial meet revision.

5 Revision on Stratified Belief Bases

In this section we will introduce one of the main contributions of the paper: the definition of a revision operator on stratified belief bases and several postulates characterizing it. This operator will allow for a multi-level reuse of discarded beliefs from the revision of each stratum.

5.1 The Revision Operator

After having presented the necessary elements to handle belief change by a set of sentences on a single belief base in Section 4 we will now define expansion and revision operators on stratified belief bases.

Definition 15 Let $\Sigma = (\Sigma_0, ..., \Sigma_n)$ be a stratified belief base and let A be a consistent set. Then the *expansion of* Σ *with respect to* A *at level* i, $0 \le i \le n$, noted by $\Sigma + (A, i)$, is equal to:

$$\Sigma' = (\Sigma_0, \ldots, \Sigma_{i-1}, \Sigma_i \cup A, \Sigma_{i+1}, \ldots, \Sigma_n)$$



Note that when $\Sigma_i' = \Sigma_i \cup A$ is inconsistent the resulting Σ' cannot be a stratified belief base because it will not satisfy the conditions of its definition, i.e., the expansion of a SBB might not be a SBB.

As stated in Section 1, now we are able to define a revision operator on stratified belief bases. The discarded beliefs from the revision of each stratum will be reused and, as a result of this, dynamically changing the value of beliefs.

Definition 16 Let $*^{j}$ be a prioritized revision operator (either kernel or partial meet) on every $\Sigma_i \in \Sigma$, for all $0 \le j \le n$. This introduces a family of revision operators of the same type: kernel or partial meet. Let A be a consistent set of beliefs, accordingly with the type, the kernel or partial meet revision of Σ with respect to A at level i, 0 < i, noted by $\Sigma * (A, i)$, is defined as:

- If i > n then $\Sigma * (A, i) = \Sigma$.
- If $i \le n$ then $\Sigma * (A, i) = \Sigma' = (\Sigma'_1, \dots, \Sigma'_m)$ where m = n or m = n + 1

 - Σ'_j = Σ_j for all j < i.
 Σ'_i = Σ_i*ⁱA and R_{i+1} = Σ_i \ Σ'_i, and then:
 - a) $\Sigma'_{i+k} = \Sigma_{i+k} *^{i+k} R_{i+k}$ and $R_{i+k+1} = \Sigma_{i+k} \setminus \Sigma'_{i+k}$ for $1 \le k \le n-i$. b) If $R_{n+1} \ne \emptyset$ then a new stratum is created as $\Sigma'_{n+1} = R_{n+1}$.

That is, our multiple revision operator can be defined by means of an iterative/recursive algorithm. We suppose that each $*^i$ is a prioritized multiple revision operator for Σ_i . The result of $\Sigma*(A,i)$ (the revision of Σ with respect to A at the level i) is $\Sigma' = (\Sigma'_1, \dots, \Sigma'_m)$, such that in general, for all j, $1 \le j \le n$:

$$\Sigma_{j}' = \Sigma_{j} *^{j} R_{j},$$

where the R_i 's are defined as follows:

- $R_j = \emptyset$ for $0 \le j < i$, so $\Sigma'_j = \Sigma_j$ for all j < i;
- $R_j = A$ for j = i;
- if $\Sigma_j \setminus \Sigma'_j \neq \emptyset$ then $R_{j+1} = \Sigma_j \setminus \Sigma'_j$.

The revised stratified belief base can have an additional layer when R_{n+1} is not empty. Since every layer Σ_i ($1 \le i \le n$) is consistent, then every subset of it is also consistent. Therefore, $R_{i+1} \subseteq \Sigma_i$ is always consistent and the revision of every layer is well defined.

We have assumed that all revision operators applied on the layers satisfy (at least) Inclusion, Consistency, Weak Success, Vacuity 1 and 2, Uniformity 1 (and 2), and Core-Retainment, according to Theorems 1 and 2. This work generalizes the proposal in [29] where only the part of the represented knowledge corresponding to undefeasible rules could be demoted transforming them in



defeasible rules; here it is possible that any sentence could be demoted and stored at a lower level. We will illustrate this revision operator below with some examples.

Note that rejection of the new information is not only justified when such information is internally inconsistent: it would seem a natural behavior that an agent when faced with (consistent) information of extremely poor quality, may reasonably opt for *flat-out rejection*, without incorporating it to its stratified belief base at any layer. An interesting aspect of this approach is that the priority given to new information is relative to its value, in that manner it is possible to incorporate it in the proper place without disturbing the information stored in higher value levels.

Example 8 This example shows two cases of revision at level 0 of a stratified belief bases Σ . First, we show the starting Σ , and then Σ' , the result of revising Σ by $A_1 = \{c\}$ at level 0. Finally, appears Σ'' , the result of revising Σ' by $A_2 = \{\neg a, \neg a \rightarrow \neg c\}$ at level 0. In each case, we are making prioritized multiple kernel revision on every layer.

Consider the set $A_1 = \{c\}$ and the stratified belief base $\Sigma = (\Sigma_0, \Sigma_1, \Sigma_2)$:

$$\Sigma_0 = \begin{Bmatrix} a \\ a \to b \\ c \to \neg b \end{Bmatrix} \Sigma_1 = \begin{Bmatrix} \neg f \\ b \to f \end{Bmatrix} \Sigma_2 = \begin{Bmatrix} e \\ e \to \neg h \\ \neg f \to h \end{Bmatrix}$$

Here follows $\Sigma' = \Sigma *^0 A_1$ showing how Σ_0 is revised by $\{c\}$. The set of $\{c\}$ -inconsistent kernels of Σ_0 is $\{\{a, a \to b, c \to \neg b\}\}$. Suppose that the consolidated incision function for Σ_0 selects $\{a \to b\}$. Therefore, Σ_1 must be revised with respect to $\{a \to b\}$. Since $\{a \to b\}$ is consistent with Σ_1 , no sentences are deleted in the revision of Σ_1 . The resulting SBB is $\Sigma' = (\Sigma'_0, \Sigma'_1, \Sigma'_2)$, where

$$\Sigma_0' = \begin{Bmatrix} a \\ c \to \neg b \\ c \end{Bmatrix} \Sigma_1' = \begin{Bmatrix} \neg f \\ b \to f \\ a \to b \end{Bmatrix} \Sigma_2' = \begin{Bmatrix} e \\ e \to \neg h \\ \neg f \to h \end{Bmatrix}$$

Sentences added to each layer are depicted in boldface.

We now revise Σ' by $A_2 = \{ \neg a, \neg a \rightarrow \neg c \}$ at level 0, i.e., $\Sigma'' = \Sigma' *^0 A_2$. The process begins by finding the set of $\{ \neg a, \neg a \rightarrow \neg c \}$ -inconsistent kernels of Σ'_0 obtaining $\{ \{a\}, \{c\} \}$. The consolidated incision function for Σ'_0 must select $\{a, c\}$. Next, the current Σ'_1 must be revised with respect to $\{a, c\}$. For this case the set of $\{a, c\}$ -inconsistent kernels of Σ'_1 is $\{ \{a \rightarrow b, b \rightarrow f, \neg f \} \}$. Assuming the consolidated incision function for Σ'_1 selects $\{\neg f\}$, the current Σ'_2 must be revised with respect to $\{\neg f\}$. In the following step, the set of $\{\neg f\}$ -inconsistent kernels of Σ'_2 contains $\{ \{e, e \rightarrow \neg h, \neg f \rightarrow h \} \}$. Suppose that the consolidated incision function for Σ'_2 selects $\{e \rightarrow \neg h\}$. Finally, a new layer



 Σ_3'' appears holding the sentence that was rejected from Σ_2'' . The resulting SBB is $\Sigma'' = (\Sigma_0'', \Sigma_1'', \Sigma_2'', \Sigma_3'')$, where

$$\Sigma_0'' = \begin{cases} c \to \neg b \\ \neg a \\ \neg a \to \neg c \end{cases} \Sigma_1'' = \begin{cases} b \to f \\ a \to b \\ a \\ c \end{cases} \Sigma_2'' = \begin{cases} \neg f \to h \\ e \\ \neg f \end{cases} \Sigma_3'' = \{ e \to \neg h \}$$

As before, sentences added to each layer are depicted in boldface.

Example 9 Consider the SBB $\Sigma = (\Sigma_0)$, where $\Sigma_0 = \{a\}$. This example shows how successive revisions at level 0 by $\{\neg a\}$, then $\{a\}$, and finally by $\{\neg a\}$ affect a simple stratified belief base.

In the example above, as the revisions are effected on Σ_0 the resulting strata contain information that can be perceived as *redundant*. Consider Σ''' , the sentence $\neg a$ is in Σ_0''' and also in Σ_2''' , and the sentence a is in Σ_1''' and also in Σ_3''' . It is possible to eliminate this redundancy by applying a process of contraction or compaction over the stratified belief base (see future works in Section 7). Nevertheless, as the system is defined here, there is no conflict with maintaining the redundancy in the stratified belief base. Observe that from Σ''' an argument $\langle \{\neg a\}, \neg a\rangle$ can be constructed, and $str_{\Sigma'''}(\{\neg a\}) = 0$. Hence, by Proposition 2, no argument for a can be obtained from a can be obtained from a can be stratified belief base. This is a general fact, if it is the case that $str_{\Sigma}(\{a\}) = i$, and $str_{\Sigma}(\{\neg a\}) = j$, i < j, the $str_{\Sigma}(\{a\}) > str_{\Sigma}(\{\neg a\})$ and therefore the sentence a will prevail.

Example 10 This example shows how stratified belief bases are revised at an arbitrary level i, with i > 0. Let us start by making a revision at level 1 by $\{f\}$, applying prioritized multiple kernel revision in cascade as necessary on every successive stratum. The followup revisions will use the sentences that are rejected in the previous stratum until the process is finished. We will show the successive states of the stratified belief base as the revisions are performed. Please notice that below we will show Σ , Σ' , Σ'' and Σ''' , but only the initial state Σ and the final state Σ''' that shows the result of the revision, should be considered as proper states; all other intermediate states are presented as an illustration of how the process progresses.

Consider the set $A_3 = \{f\}$ and the stratified belief base $\Sigma = (\Sigma_0, \Sigma_1, \Sigma_2)$:

$$\Sigma_{0} = \begin{cases} a \\ a \to b \\ c \to \neg b \end{cases} \Sigma_{1} = \begin{Bmatrix} \neg f \\ b \to f \end{Bmatrix} \Sigma_{2} = \begin{Bmatrix} e \\ e \to \neg h \\ \neg f \to h \end{Bmatrix}$$



If we revise Σ_1 by $\{f\}$ we obtain Σ' , i.e., $\Sigma' = \Sigma *^1 A_3$. To perform this revision we consider the set of $\{f\}$ -inconsistent kernels of Σ_1 , that in this case is $\{\{\neg f\}\}$. The only possible consolidated incision function for Σ_1 must select $\{\neg f\}$. This leads to a revision of Σ_2 with respect to $\{\neg f\}$, obtaining the first intermediate state $\Sigma' = (\Sigma'_0, \Sigma'_1, \Sigma'_2)$:

$$\Sigma_0' = \begin{cases} a \\ a \to b \\ c \to \neg b \end{cases} \Sigma_1' = \begin{cases} b \to f \\ f \end{cases} \Sigma_2' = \begin{cases} e \\ e \to \neg h \\ \neg f \to h \end{cases}$$

Again, sentences added to each layer are depicted in boldface.

Now it is necessary to revise Σ_2'' by $\{\neg f\}$, i.e., $\Sigma'' = \Sigma' *^2 \{\neg f\}$. In this case the set of $\{\neg f\}$ -inconsistent kernels of Σ_2' is $\{\{e, e \rightarrow \neg h, \neg f \rightarrow h\}\}$. Suppose for the sake of the example that the consolidated incision function for Σ_2' selects $\{\neg f \rightarrow h\}$. We obtain the next intermediate state $\Sigma'' = (\Sigma_0'', \Sigma_1'', \Sigma_2'')$, where:

$$\Sigma_0'' = \begin{cases} a \\ a \to b \\ c \to \neg b \end{cases} \Sigma_1'' = \begin{cases} b \to f \\ f \end{cases} \Sigma_2'' = \begin{cases} e \\ e \to \neg h \end{cases}$$

The sentence $\{\neg f \rightarrow h\}$ should be used to revise Σ_3' , that will be created empty, i.e., the new layer Σ_3''' will hold the sentence $\{\neg f \rightarrow h\}$ that was rejected from Σ_2'' , leading to the final result $\Sigma'' = (\Sigma_0''', \Sigma_1''', \Sigma_2''', \Sigma_3''')$, where:

$$\Sigma_0^{\prime\prime\prime} = \begin{cases} a \\ a \to b \\ c \to \neg b \end{cases} \Sigma_1^{\prime\prime\prime} = \begin{cases} b \to f \\ f \end{cases} \Sigma_2^{\prime\prime\prime} = \begin{cases} e \\ e \to \neg h \end{cases} \Sigma_3^{\prime\prime\prime} = \{ \neg f \to h \}$$

Note that when Σ is revised by the new information $\{f\}$ at level 1, causes $\{\neg f\}$ to be demoted to the level 2, which in turn forces some of the beliefs at the level 2 to be demoted to the level 3. In this case, the revision process does not affect the beliefs at level 0 because the new information used to revise Σ was not good enough to cast doubt on that level.

Revision operators for stratified belief bases show a sort of *cascade effect*, in which information demoted from stratum i to layer i+1 is preserved in Σ_{i+1} and, when necessary, some other information contained in this stratum can be further demoted to avoid inconsistency. We handle each demotion at stratum i as a revision by an information of value i. We adopt this criteria as a generalization of the *one-level demotion* used in [29], trying to expand the argumentative inferences in this framework.

5.2 Postulates for Revision on SBBS

From the construction, and following some accepted rationality criteria, we propose a set of postulates for the new revision operator introduced in Definition 16. Let $\Sigma = (\Sigma_0, \dots, \Sigma_n)$ be a stratified belief base and A be a consistent set of sentences. Let $\Sigma' = (\Sigma'_0, \dots, \Sigma'_m)$ with m = n or m = n + 1 be the stratified belief base which results from the revision of Σ at the level i



with respect to A, noted by $\Sigma *(A, i)$. We propose the following postulates for revision:

SBB-Conservativity: $\Sigma*(A,i)$ is a stratified belief base. SBB-Success: $A\subseteq\Sigma_i'$ where $\Sigma_i'\in\Sigma'=\Sigma*(A,i)$. SBB-Vacuity 1: If $A\subseteq\Sigma_i, \Sigma_i\in\Sigma$ then $\Sigma*(A,i)=\Sigma$. SBB-Vacuity 2: If $A\cup\Sigma_i\not\vdash\bot$ then $\Sigma*(A,i)=\Sigma+(A,i)$. SBB-Consistency: For all $\Sigma_i'\in\Sigma'$ it holds that Σ_i' is consistent.

SBB-Inclusion: $\bigcup (\Sigma * (A, i)) \subseteq \bigcup (\Sigma + (A, i)).$ SBB-Strengthening: If $\Sigma' = \Sigma * (A, i)$ then $str_{\Sigma'}(A) \leq i.$

SBB-Weakening: Given $\Sigma_i \in \Sigma$ and $\Sigma_i' \in \Sigma * (A, i)$, if $\alpha \in \Sigma_i$ and $\alpha \notin \Sigma_i'$ then

it holds that $\alpha \in \Sigma'_{i+1}$.

SBB-Conservativity establishes that the revision of a stratified belief base is a (possibly new) stratified belief base. SBB-Success determines that the revision is prioritized since the input set is included in the stratum i of the revised stratified belief base. This approach incorporates a notion of success far more articulated and less counterintuitive than most other versions present in the belief revision literature: nothing is changed in the strata of higher value. SBB-Vacuity 1 says that nothing changes when the input set is already included in the stratum i. sbb-Vacuity 2 says that nothing is removed when the input set is consistent with the stratum i. SBB-Consistency determines that every stratum of the revised stratified belief base is consistent. SBB-Consistency is implied by SBB-Conservativity and the definition of stratified belief bases. SBB-Inclusion establishes that the union of strata in the revision of Σ by A is always included in the union of strata in the expansion of Σ by A. However, observe that sbb-Inclusion does not imply the inclusion among strata: if $\Sigma = (\Sigma_0, \dots, \Sigma_n)$ and $\Sigma' = \Sigma * (A, i) = (\Sigma'_0, \dots, \Sigma'_m)$ with m = n or m = n + 1 then it is not ensured that $\Sigma_i \subseteq \Sigma_i'$ for all $1 \le i \le n$. Finally, sbb-Strengthening and sbb-Weakening show how the value of beliefs may change after a revision process.

Proposition 3 Let * be a multiple revision operator at the level i defined according to Definition 16. Then * satisfies sbb-Conservativity, sbb-Success, sbb-Vacuity 1, sbb-Vacuity 2, sbb-Inclusion, sbb-Strengthening, and sbb-Weakening.

Proof

SBB-Conservativity: $\Sigma * (A, i)$ is a stratified belief base.

We must show that every stratum $\Sigma_i' \in \Sigma * (A, i)$ is consistent for all $0 \le i \le m$. For $0 \le i \le n$, we have two cases:

- $\Sigma_i' = \Sigma_i$. Straightforward because Σ is a stratified belief base.
- $\Sigma_i' = \Sigma_i *^i R_i$. Then, by $*^i$ satisfying *consistency*, it holds that Σ_i' is consistent.

If m = n we are done. If m = n + 1 then $R_{n+1} \neq \emptyset$. Then, by Definition 16, $\Sigma'_m = R_{n+1}$ and, since Σ_n is consistent,



then every subset of Σ_n is consistent. Since $R_{n+1} \subseteq \Sigma_n$ then Σ'_m is consistent and we are done.

SBB-Success: $A \subseteq \Sigma_i' \in \Sigma' = \Sigma * (A, i).$

Straightforward due to $*^i$ satisfying weak success.

SBB-Vacuity 1: If $A \subseteq \Sigma_i \in \Sigma$ then $\Sigma * (A, i) = \Sigma$.

If $A \subseteq \Sigma_i \in \Sigma$ then A is consistent. Then, by $*^i$ satisfying vacuity 2, it holds that $\Sigma_i *^i A = \Sigma_i \cup A = \Sigma_i$. Therefore, $\Sigma'_i = \Sigma_i$ for all $0 \le i \le n$ and therefore $\Sigma *(A, i) = \Sigma$.

SBB-Vacuity 2: If $A \cup \Sigma_i \nvdash \bot$ then $\Sigma * (A, i) = \Sigma + (A, i)$.

If $A \cup \Sigma_i \nvdash \bot$ then, by $*^i$ satisfying *vacuity* 2, it holds that $\Sigma'_j = \Sigma_i *^i A = \Sigma_i \cup A$. By Definition 16, $\Sigma'_j = \Sigma_i$ for all $j \neq i$. Therefore, $\Sigma * (A, i) = \Sigma + (A, i)$.

sвв-Consistency It is already implied by sвв-Conservativity.

SBB-Inclusion: $| \int (\Sigma * (A, i)) \subseteq | \int (\Sigma + (A, i)).$

Straightforward from Definition 16 and *' satisfying

inclusion.

SBB-Strengthening: $str_{\Sigma*(A,i)}(A) \leq i$.

By $*^i$ satisfying *weak success* it holds that $A \subseteq \Sigma_i' \in \Sigma * (A, i)$. By Definition 5, the strength of the set $A = \{x_1, \ldots, x_s\}$ is:

 $str_{\Sigma*(A,i)}(A) = \max\{str_{\Sigma*(A,i)}(x_1), \dots, str_{\Sigma*(A,i)}(x_s)\}\$

By Definition 4, $str_{\Sigma*(A,i)}(x) = j$ if and only if $x \in \Sigma'_j$ and there is no Σ'_k (k < j) such that $x \in \Sigma'_k$. Since the revision ensures that all $x_l \in A$ are in Σ'_i but also may be in more valued strata, then $str_{\Sigma*(A,i)}(A) \le i$.

SBB-Weakening: If $\alpha \in \Sigma_i \in \Sigma$ and $\alpha \notin \Sigma_i' \in \Sigma * (A, i)$ then it holds that

 $\alpha \in \Sigma'_{i+1} \in \Sigma * (A, i)$.

Suppose that $\alpha \in \Sigma_i \in \Sigma$ and $\alpha \notin \Sigma_i' \in \Sigma * (A, i)$. Then $\Sigma_i \neq \Sigma_i'$ and $\Sigma_i' = \Sigma_i *^i A$. Hence $\alpha \in R_{i+1} = \Sigma_i \setminus \Sigma_i'$. By Definition 16, $\Sigma_{i+1}' = \Sigma_{i+1} *^{i+1} R_{i+1}$ and, by $*^{i+1}$ satisfying weak success, it holds that $R_{i+1} \subseteq \Sigma_{i+1}' = \Sigma_{i+1} *^{i+1} R_{i+1}$.

Therefore $\alpha \in \Sigma'_{i+1} \in \Sigma * (A, i)$.

Proposition 3 shows that the well-behavedness of the revision operators used on each stratum induces a well-behavedness on the revision operator for stratified belief bases in a straightforward way. In the proposed stratified belief base revision, new information can be taken considering different values, and the reuse of sentences is multi-level, something very interesting for the modeling of dynamics in stratified belief bases. This point addresses a novel issue in belief revision. Such changes are not handled properly by modifying the retraction and addition of beliefs. The basic idea is that inconsistencies arising when new information is incorporated into the stock of beliefs can be eliminated not only by removing beliefs, but also by weakening them, i.e., storing them in a lower value level.

Stratified Belief Bases Revision with Argumentative Inference

5.3 Argumentation and Belief Revision

The possibility of changing the value of defeasible beliefs endows revision operations with a new quality with some important consequences for the corresponding argumentation systems. Indeed, a connection between belief revision and argumentation is immediately established by considering which changes in the argumentation structures are caused by the revision of the stratified belief base that gives rise to these arguments. If A is an argument for some belief α , then when the revision $\Sigma*(A,i)$ is effected leading to Σ' from which a new set of arguments can be created and a new defeat relation among the arguments can be defined. The following examples will help to clarify these comments.

Example 11 Consider the following stratified belief base Σ :

$$\Sigma_0 = \begin{cases} a \\ \neg h \\ c \\ a \rightarrow b \end{cases} \qquad \Sigma_1 = \{c \rightarrow \neg b\}$$

Here, we can obtain the argument $\langle A_1, b \rangle$ where $A_1 = \{a, a \rightarrow b\}$ with $str_{\Sigma}(\langle A_1, b \rangle) = 0$. Note that no argument for " $\neg b$ " can be obtained (see Proposition 2), and b is warranted. Consider now that the stratified belief base is revised with respect to the set " $\{a \rightarrow b\}$ " at the level 0. Suppose that we will make a prioritized multiple kernel revision of every stratum. Then, the set of $\{a \rightarrow b\}$ -inconsistent kernel of Σ_0 is $\{\{a, \neg b\}\}$. Suppose that the consolidated incision function for Σ_0 selects $\{a\}$. Then Σ_1 must be revised with respect to $\{a\}$. Since $\{a\}$ is consistent with Σ_1 then no sentences are deleted in the respective revised stratum. Let $\Sigma' = \Sigma * (\{a \rightarrow b\}, 0)$. The result of the revision is:

$$\Sigma_0' = \begin{cases} \neg h \\ c \\ a \to b \\ a \to h \end{cases} \qquad \Sigma_1' = \begin{cases} c \to \neg b \\ a \end{cases}$$

Observe that sentence "a" is now in Σ'_1 and therefore, the value of the argument $\langle A_1, b \rangle$ changes. As another consequence of this change, from Σ' an argument for " $\neg b$ " can now be obtained, and therefore, b is not warranted.

$$\langle A_1, b \rangle$$
 $A_1 = \{a, a \rightarrow b\}$ $str_{\Sigma'}(A_1) = 1$
 $\langle A_2, \neg b \rangle$ $A_2 = \{c, c \rightarrow \neg b\}$ $str_{\Sigma'}(A_2) = 1$

Example 12 Consider the stratified belief base Σ' :

$$\Sigma_0' = \{e\} \quad \Sigma_1' = \left\{ \begin{matrix} \neg h \\ c \\ a \to b \\ a \to h \end{matrix} \right\} \quad \Sigma_2' = \left\{ \begin{matrix} c \to \neg b \\ a \end{matrix} \right\}$$



Here it holds that:

$$\langle A_1, b \rangle$$
 $A_1 = \{a, a \rightarrow b\}$ $str_{\Sigma'}(A_1) = 2$
 $\langle A_2, \neg b \rangle$ $A_2 = \{c, c \rightarrow \neg b\}$ $str_{\Sigma'}(A_2) = 2$

Consider now that Σ' is revised with respect to the set " $\{\neg a, \neg b, \neg c\}$ " at the level 1. Again, suppose that we will make a prioritized multiple kernel revision of every stratum. Then, the set of $\{\neg a, \neg b, \neg c\}$ -inconsistent kernels of Σ'_1 is $\{\{a \rightarrow b\}, \{c\}\}$. Since the consolidated incision function must select at least one sentence of each set of the above set (of sets), then it selects $\{a \rightarrow b, c\}$.

In the next step, Σ' will be revised with respect to the set " $\{a \to b, c\}$ " at the level 2. Then, the set of $\{a \to b, c\}$ -inconsistent kernels of Σ'_2 is equal to $\{\{c \to \neg b, a\}\}$. Suppose that the consolidated incision function for Σ'_2 selects $\{a\}$. Since Σ'_3 does not exists, then $\{a\}$ will be the new stratum of the revised stratified belief base. Let $\Sigma'' = \Sigma'*(\{\neg a, \neg b, \neg c\}, 1)$. The result of the revision is:

$$\Sigma_{0} = \{e\} \qquad \Sigma_{1}^{"} = \begin{cases} \neg h \\ a \to h \\ \neg a \\ \neg b \\ \neg c \end{cases} \qquad \Sigma_{2}^{"} = \{c \\ a \to b \\ c \to \neg b \end{cases} \qquad \Sigma_{3}^{"} = \{a\}$$

Observe that sentence "a" is now in Σ_3 and therefore, the value of the arguments change.

$$\langle A_1, b \rangle$$
 $A_1 = \{a, a \rightarrow b\}$ $str_{\Sigma'}(A_1) = 3$
 $\langle A_2, \neg b \rangle$ $A_2 = \{c, c \rightarrow \neg b\}$ $str_{\Sigma'}(A_2) = 2$

6 Related Work

In this section we will give a brief summary of some related works in which stratified belief bases, belief revision and argumentation are studied (see also [30]). We start with comparing out approach to a paper that seems most closely related.

The research presented in [7] also sets up a framework for argumentative inferences from stratified belief bases, where the semantics of the different strata are to reflect reliability, and the modeling is based on possibility degrees. More precisely, knowledge bases are stratified, namely each formula in the knowledge base is associated with its level of certainty, or reliability, corresponding to the layer to which it belongs. At first sight, the way they define the derivation of conclusions, called π -consequence, via arguments presents some commonalities with our approach. However, the evaluation of arguments sanctioning the inference is substantially simpler, as only conflicts between heads of arguments are considered; the definition of conflict does not consider



sub-arguments, i.e., just conflicts between conclusions are recognized. This may lead to consequences that are against intuition. We will clarify this using an example taken from [32] adapting it to our framework.

Example 13 The agent believes (with different degrees of reliability) that chicken and penguins are birds $(c \to b, p \to b)$, that both species do not fly $(c \to \neg f, p \to \neg f)$ although birds usually fly $(b \to f)$, but the agent believes that scared chicken may fly $(cs \to f)$. Moreover, it believes that flying things usually nest in trees $(f \to n)$. At the moment, it is observing a scared chicken (c, s). The beliefs of the agent are encoded in the following stratified belief base:

$$\Sigma_{0} = \begin{cases} c \\ s \\ c \to b \\ p \to b \end{cases} \qquad \Sigma_{1} = \begin{cases} p \to \neg f \\ cs \to f \end{cases} \qquad \Sigma_{2} = \begin{cases} c \to \neg f \\ f \to n \end{cases} \qquad \Sigma_{3} = \{b \to f\}$$

With $A_1 = \{c, c \to \neg f\}$, $A_2 = \{c, c \to b, b \to f, f \to n\}$, the arguments $\langle A_1, \neg f \rangle_{\Sigma}$ and $\langle A_2, n \rangle_{\Sigma}$ can be built. In our approach, $\langle A_1, \neg f \rangle_{\Sigma}$ defeats $\langle A_2, n \rangle_{\Sigma}$ at the sub-argument $\langle \{c, c \to b, b \to f\}, f \rangle_{\Sigma}$ and so inhibits the derivation of n. The argumentative inference of [7], however, derives n, since there is no argument for $\neg n$.

Thus, our notion of defeat among arguments is substantially more complex than theirs. When the arguments are built and the defeat relation is established, the dialectical evaluation of the status of these arguments is carried out applying one of the possible Dung's style semantics.

The second essential difference with the work Benferhat and colleagues is that we revise the stratified belief base to ensure each stratum to be consistent. The research introduced in [7], and in a closely related work [8], basically aim at treating inconsistencies caused by the use of multiple sources of information, and they use a restricted form of argumentation for paraconsistent reasoning. They suggest that it is not even necessary to restore consistency on the level of the strata in order to make sensible inferences from an inconsistent knowledge base. The problem with adopting that view is that adding more inconsistent beliefs in the same stratum would yield more contradictory arguments of equal strength, so less conclusions can be supported effectively. This would correspond to an agent which gets increasingly more confused by new information that it is incorporated just superficially. In our approach, conflicts caused by new information are resolved by revision, equipping the agent with a consistent set of beliefs at each level, i.e., at any layer of the stratified belief base from which it may obtain informative conclusions via argumentation.

In a recent work [9], an interesting and useful family of computationally effective strategies for conflict resolution which can be used for iterated belief revision and merging information from multiple sources is developed. The resulting strategies are favorably compared and contrasted with other



existing approaches to the problem of handling inconsistent information. The information is assumed to be given as ordered consistent knowledge bases, i.e., a set of ranked information represented as logical sentences. The origins of the research can be traced to [61] where an Adjustment procedure is presented; this is a system able to deal with conflicts in knowledge bases. These conflicts are produced as new information is received possibly leading to inconsistencies. The procedure propagates as many highly ranked formulas as possible, ignoring information at the highest rank where an inconsistency is found and information below that rank. The main disadvantage of Adjustment is that, to restore the consistency after accepting new information in a context where the independence of information is not made explicit, it can eliminate more content of the knowledge base than it is necessary. An improvement on this procedure is another strategy called Maxi-Adjustment [62]. Maxi-Adjustment solves conflicts at each rank of priority in the knowledge base in an incremental manner, starting from the information with highest rank. When inconsistency is encountered in the knowledge base, all formulas in the rank responsible for the conflicts are removed. The other formulas are kept, and the process continues to the next rank. The approach used in [9] involves the weakening of the information involved in conflicts instead of discarding it, minimizing in that way the loss of information; this approach is called Disjunctive Maxi-Adjustment (DMA). DMA improves Maxi-Adjustment by solving the conflicts at each rank of priority in the knowledge base and doing that in an incremental manner, starting from the information with highest rank. The full description of this strategy is beyond the reach of this paper, suffices to say that this is a pure belief revision procedure that re-establishes consistency. Although this weakening process superficially reminds our approach in its goal of comply with the Minimal Change principle, their research unlike ours does keep a consistent knowledge base. By introducing an argumentation reasoning procedure, although we keep each stratum (or rank in their terminology) consistent the complete set of strata stores conflicting information. We solve these conflicts by using the aforementioned argumentation reasoner. In our view, these setup permits to keep more information since the conflicts could be solved by demoting.

In [15] preferred subtheories as a generalization of default theories are introduced. Basically, preferred subtheories are maximal consistent subsets of stratified belief bases, and arguments are built from them to support conclusions. Brewka studied revision of stratified belief bases in a further paper [14]. However, while Brewka realizes revision by expanding the corresponding layer and making use of the underlying nonmonotonic reasoning machinery, our approach aims at resolving conflicts between annotated beliefs by applying belief base revision operators and argumentation theory.

The research introduced in [16] proposes a way to deal with contradicting information by considering a support ordering on that information. For Cantwell, information received from different sources could be inconsistent. Even when the sources of information can be ordered on the basis of their trustworthiness, it turns out that extracting an acceptable notion of support



for information is a non-trivial matter, as is the question what information a rational agent should accept. This approach is shown to be closely related to the epistemic entrenchment and Grove spheres studied in the belief revision community.

The dynamics of a belief revision system was studied in [53]. They considered the relations among beliefs from a "derivational approach" trying to derive a theory of belief revision from a more concrete epistemological theory. According to them, one of the goals of belief revision is to generate a knowledge base in which each piece of information is justified (by perception) or warranted by arguments containing previously held beliefs. The difficulty is that the set of justified beliefs can exhibit all kinds of logical incoherences because it represents an intermediate stage in reasoning. Therefore, they propose a theory of belief revision concerned with warrant rather than justification.

In a framework oriented to defeasible reasoning, Falappa et al. [29] proposed a non-prioritized revision operator based on the use of explanations (arguments). In this formalism epistemic states are represented as a set of undefeasible beliefs (called strict knowledge) and defeasible beliefs that support tentative (and potentially inconsistent) conclusions. The idea is that an agent, before incorporating information which is inconsistent with its knowledge, requests an explanation supporting it. One of the most interesting ideas of this work is the generation of defeasible conditionals from a revision process. This approach preserves consistency in the undefeasible knowledge and it provides a mechanism to dynamically qualify the beliefs as undefeasible or defeasible.

A theory to cope with the dynamics of possibly inconsistent information represented by a finite state is presented in [59]. Since the agent's information will always be finite, Tamminga proposes the use of finite valuations to represent it, ensuring economy of representation and avoiding redundant valuations. Finite operations for expansion and contractions are defined, without the presence of extra-logical information such as choice functions or an ordering over (sets of) sentences.

In a research that puts together argumentation and belief revision in the same conceptual framework, the important role played by Toulmin's layout of argument in fostering such integration is highlighted [50]. They consider argumentation as 'persuasion to believe' and this restriction is useful to make more explicit the connection with belief revision. They propose a model of belief dynamics alternative to the AGM approach: Data-oriented Belief Revision (DBR). Two basic informational categories, data and beliefs, are put forward in their model, to account for the distinction between pieces of information that are simply gathered and stored by the agent (data), and pieces of information that the agent considers (possibly up to a certain degree) truthful representations of states of the world (beliefs). Whenever a new piece of evidence is acquired through perception or communication, it affects directly the agent's data structure and only indirectly his beliefs. Belief revision is often triggered by information update either on a fact or on a source: the agent receives a new piece of information, rearranges his data structure accordingly, and possibly changes his beliefs.



A direct relationship between argumentation and belief revision is shown in [12]. They consider argumentation as persuasion to believe and that persuasion should be related to belief revision. More recently, Boella et al. [13] presented the interrelation between argumentation and belief revision on Multiagent Systems. When an agent uses an argument to persuade another one, he must consider not only the proposition supported by the argument, but also the overall impact of the argument on the beliefs of the addressee.

In [56] an abstract theory that captures the dynamics of an argumentation framework through the application of belief revision concepts is introduced. They define a dynamic abstract argumentation theory including dialectical constraints, and then present argument revision techniques to describe the fluctuation of the set of active arguments (the ones considered by the inference process of the theory). They claim that the theory should be abstract from two standpoints: (1) there is no restriction to any particular representation for arguments nor argumentative semantics, (2) the characterization of the change operators (specially contractions), which is not restricted to a particular implementation. They define expansion, contraction, and revision operators, where the latter can be expressed in terms of the other two, leading to an identity similar to the one defined by Isaac Levi [45]. Their abstract theory allows the introduction of an argument ensuring it can be believed afterwards. This is achieved by applying a revision, that is, an expansion followed by a contraction. The expansion operator is quite straightforward, but (as usual in any model for the theory change) the main complexity relies on the definition of the contraction operator, which allows a wide range of possibilities: from affecting unrestrictedly any number of arguments in the system to keeping this perturbation to a minimum. This choice is up to the minimal change principle followed by the specification of the contraction operation, which also has an indirect impact over the defeat relation among arguments. Moguillansky et al. [48, 49] proposed an instantiation of their operators to Defeasible Logic Programming, DeLP [32]. The main contribution is to define an argument revision operator that ensures warrant of the conclusion of the (external) argument being added to a defeasible logic program. When they revise a program by an argument $\langle A, \alpha \rangle$ (where A is an argument for α), the program resulting from the revision will be such that A is an undefeated argument and α is therefore warranted.

In a recent work [17], a proposal for handling the addition of an argument to a Dung-style abstract argumentation system was introduced. Several change operations were defined considering four possibilities: modifying the attack relation (in Dung's sense) by adding or removing an attack, adding an argument together with a set of interactions, and removing an argument together with all interactions related to it, where an interaction is an attack received or effected by the argument involved. They define an argumentation framework $\langle \mathbb{A}, \mathbb{R} \rangle$ where \mathbb{A} is a finite set of arguments and \mathbb{R} is an attack relation among arguments. The main issue of any argumentation system is the selection of acceptable sets of arguments. An acceptable set of arguments must be under certain criteria coherent. An argumentation semantic defines the properties



required for a set of arguments to be acceptable. The selected sets of arguments under a given semantics are called *extensions* of that semantic. According to [17], the outcome of a revision process is the set of extensions under a given semantic. Then, by considering how the set of extensions is modified under the revision process, they propose a typology of different revisions: decisive revision and expansive revision. Narrowing the research to two particular semantics: the grounded and preferred semantics [26], the impact on the set of extensions that such changes effect on the outcome of the argumentation was described.

In a logic programming setup, the work by Delgrande et al. [20] addresses the problem of belief revision in (nonmonotonic) logic programming under answer set semantics: given logic programs P and Q, the goal is to determine a program R that corresponds to the revision of P by Q, denoted $P \ast Q$. They proposed formal techniques analogous to those of distance-based belief revision in propositional logic. They investigate two specific operators: (logic program) expansion and a revision operator based on the distance between the SE models of logic programs.

7 Conclusions and Future Work

In this work we have introduced a revision operator on stratified belief bases characterizing it through a set of postulates. The information removed from the revised stratum is reused to revise the next less valued stratum, and so on. This revision mechanism brings about the possibility of changing the belief's value. The reuse of the removed beliefs improves de possibility of satisfying the principle of minimal change as proposed by the belief change theory in general and in the AGM model in particular. Then, by using a formal framework for argumentation, we exemplified how after a revision process the status of arguments is modified leading to changes in the set of inferred beliefs. As future work, we are working in stating representation theorems for this new type of revision over stratified belief bases.

In this work we have not explored the possibility of *promoting* pieces of information (beliefs), i.e., moving it from one layer to another with higher value. The operator was defined in such a way that beliefs can only be *demoted* after being eliminated from a layer in the process of restoring consistency; that belief can only be reinserted in a higher layer through a new addition from external sources. However, we hypothesize that the *promotion* of beliefs could be prompted in other forms diverse from the ways demotion is; it is possible that beliefs can be promoted for *internal* reasons regardless of the arrival of new external information. We are exploring the characterization of a more complete framework, with dual operators for the promotion and demotion of beliefs.

Another line of work in this research will be the development of a more extended set of change operators. An operation will be a *one-level contraction*, taking a stratified belief base Σ , a set A, and a stratum i, and eliminating all



sentences of A from the stratum i of Σ . Another operation will be a *multi-level* contraction which takes a stratified belief base Σ , and a set A, and generates that all sentences of A are removed from each stratum of Σ . The last operation will be a *compaction* which takes a stratified belief base Σ and generates a new stratified belief base Σ' such that there is no strata i and j and no sentence α such that $i \neq j$, $\alpha \in \Sigma'_i$ and $\alpha \in \Sigma'_j$.

In some recent developments in argumentation theory, several ways of assigning strength to each argument have been introduced resulting in interesting argumentation frameworks. Using this results, and understanding strength as value, we can consider that many relatively weaker arguments may "accumulate" or "accrue" their relative strength overcoming a single argument that is stronger that any one of them individually [36, 37, 54, 60]. For instance, if we have one argument for α with a given strength and we have a set of arguments for $\neg \alpha$ with lower individual strength, we could consider the idea that all them working together could defeat the stronger argument. That is, when a number of (independent) reasons against a given conclusion accrue, this should have some effect on the strength of that conclusion, even when each of these reasons taken individually would not have the strength sufficient to produce such effect. This idea could also be used to promote information considered stronger when a set of arguments supports it.

Acknowledgements We would like to thank to the anonymous referees for very valuable comments on earlier versions of this paper. This work is partially supported by Universidad Nacional del Sur (UNS), Consejo Nacional de Investigaciones Científicas y Técnicas (CONICET) and Ministerio de Ciencia y Tecnología (MinCyT) from Argentina, and Deutscher Akademischer Austausch Dienst (DAAD) from Germany.

References

- Alchourrón, C., Gärdenfors, P., & Makinson, D. (1985). On the logic of theory change: Partial meet contraction and revision functions. *The Journal of Symbolic Logic*, 50, 510–530.
- Alchourrón, C., & Makinson, D. (1985). On the logic of theory change: Safe contraction. Studia Logica, 44, 405–422.
- Baroni, P., & Giacomin, M. (2007). Comparing argumentation semantics with respect to skepticism. In 9th European Conference on Symbolic and Quantitative Approaches to Reasoning with Uncertainty, (ECSQARU 2007) (pp. 210–221). Tunisia: Hammamet.
- 4. Baroni, P., & Giacomin, M. (2009). Semantics of abstract argument systems. In I. Rahwan, & G. R. Simari (Eds.), *Argumentation in Artificial Intelligence*, (pp. 24–44). Springer.
- 5. Baroni, P., & Giacomin, M. (2009). Skepticism relations for comparing argumentation semantics. *International Journal Approximate Reasoning*, 50(6), 854–866.
- Bench-Capon, T. J. M., & Dunne, P. E. (2007). Argumentation in artificial intelligence. Artificial Intelligence, 171(10–15), 619–641.
- 7. Benferhat, S., Dubois, D., & Prade, H. (1993). Argumentative inference in uncertain and inconsistent knowledge bases. In D. Heckerman, & E. H. Mamdani (Eds.), *Proceedings of 9th*

³Note that we are formulating a *package contraction*, where all the sentences of the epistemic input must be removed from the epistemic state. The other alternative is to define a *choice contraction*, where it is sufficient that at least one of the sentences is removed. For more details about package and choice contraction, see [31].



- Annual Conference on Uncertainty in Artificial Intelligence, UAI'1993 (pp. 411–419). Morgan Kaufmann.
- 8. Benferhat, S., Dubois, D., & Prade, H. (1995). How to infer from inconsistent beliefs without revising. In *Proceedings of IJCAI'1995* (pp. 1449–1455).
- 9. Benferhat, S., Kaci, S., Berre, D. L., & Williams, M.-A. (2004). Weakening conflicting information for iterated revision and knowledge integration. *Artificial Intelligence Journal*, 153(1–2), 339–371.
- 10. Besnard, P., & Hunter, A. (2001). A logic-based theory of deductive arguments. *Artificial Intelligence*, 128(1–2), 203–235.
- 11. Besnard, P., & Hunter, A. (2009). Argumentation based on classical logic. In I. Rahwan, & G. R. Simari (Eds.), *Argumentation in Artificial Intelligence*, (pp. 133–152). Springer.
- 12. Boella, G., Costa Perera, C. d., Tettamanzi, A., & van der Torre, L. (2008). Dung argumentation and AGM belief revision. In 5th International Workshop on Argumentation in Multi-Agent Systems, ArgMAS'2008.
- 13. Boella, G., Costa Perera, C. d., Tettamanzi, A., & van der Torre, L. (2008). Making others believe what they want. *Artificial Intelligence in Theory and Practice II* (pp. 215–224).
- 14. Brewka, G. (1989). Belief revision in a framework for default reasoning. In A. Fuhrmann, & M. Morreau (Eds.), *The logic of theory change. Lecture Notes in Computer Science* (Vol. 465, pp. 206–222). Springer.
- Brewka, G. (1989). Preferred subtheories: An extended logical framework for default reasoning. In *Proceedings 11th Joint Conference on Artificial Intelligence, IJCAI 1989* (Vol. 2, pp. 1043–1048). San Mateo, California.
- Cantwell, J. (1998). Resolving conflicting information. *Journal of Logic, Language and Infor*mation, 7(2), 191–220.
- 17. Cayrol, C., de Saint Cyr, F. D., & Lagasquie Schiex, M. C. (2008). Revision of an argumentation system. In *Proceedings of the International Conference on Principles of Knowledge Representation and Reasoning, KR 2008* (pp. 124–134).
- 18. Darwiche, A., & Pearl, J. (1997). On the logic of iterated belief revision. *Artificial Intelligence*, 89, 1–29.
- 19. Delgrande, J., Dubois, D., & Lang, J. (2006). Iterated revision and prioritized merging. In *Proceedings of KR 2006* (pp. 210–220). AAAI Press.
- 20. Delgrande, J., Schaub, T., Tompits, H., & Woltran, S. (2008). Belief revision of logic programs under answer set semantics. In *Proceedings of the 11th International Conference on Principles of Knowledge Representation and Reasoning* (pp. 411–421).
- 21. Delgrande, J. P., Schaub, T., Tompits, H., & Woltran, S. (2008). Belief revision of logic programs under answer set semantics. In G. Brewka, & J. Lang (Eds.), *KR* (pp. 411–421). AAAI Press.
- 22. Doyle, J. (1979). A truth maintenance system. Artificial Intelligence, 12, 231–272.
- Doyle, J. (1992). Reason maintenance and belief revision: Foundations versus coherence theories. In P. Gärdenfors (Ed.), *Belief revision*, (pp. 29–51). Cambridge University Press.
- Dubois, D. (2006). Three scenarios for the revision of epistemic states. In *Proceedings of NMR*'2006.
- 25. Dung, P. M. (1995). On the acceptability of arguments and its fundamental role in non-monotonic reasoning, logic programming and n-person games. *Artificial Intelligence*, 77(2), 321–358.
- Dung, P. M. (1995). On the acceptability of arguments and its fundamental role in nonmonotonic reasoning, logic programming and n-person games. *Artificial Intelligence*, 77, 321– 357.
- Falappa, M. A., Fermé, E. L., & Kern-Isberner, G. (2006). On the logic of theory change: Relations between incision and selection functions. In *Proceedings of ECAI-2006* (pp. 402–406).
- Falappa, M. A., Kern-Isberner, G., Reis, M. D. L., & Simari, G. R. (2011). Prioritized and non-prioritized multiple change on belief bases. *The Journal of Philosophical Logic*. doi:10.1007/s10992-011-9200-8
- 29. Falappa, M. A., Kern-Isberner, G., & Simari, G. R. (2002). Belief revision, explanations and defeasible reasoning. *Artificial Intelligence Journal*, 141, 1–28.



- Falappa, M. A., Kern-Isberner, G., & Simari, G. R. (2009) Argumentation in Artificial Intelligence. In I. Rahwan, & G. R. Simari (Eds.), *Belief Revision and Argumentation Theory* (pp. 341–360). Springer.
- 31. Fuhrmann, A., & Hansson, S. O. (1994). A survey of multiple contractions. *The Journal of Logic, Language and Information*, *3*, 39–76.
- 32. García, A. J., & Simari, G. R. (2004). Defeasible logic programming: An argumentative approach. *Theory and Practice of Logic Programming*, 4(1), 95–138.
- 33. Gärdenfors, P. (1988). *Knowledge in flux: Modelling the dynamics of epistemic states*. The MIT Press, Bradford Books: Cambridge.
- 34. Gärdenfors, P. (1990). The dynamics of belief systems: Foundations vs. coherence theories. *Revue Internationale of Philosophie*, 44, 24–46.
- 35. Gärdenfors, P., & Makinson, D. (1988). Revisions of knowledge systems using epistemic entrenchment. 2nd conference on theoretical aspects of reasoning about knowledge (pp. 83–95).
- Gómez Lucero, M. J., Chesñevar, C. I., & Simari, G. R. (2009). Modelling argument accrual in possibilistic defeasible logic programming. In *Proceedings of ECSQARU*'2009 (pp. 131–143).
- 37. Gómez Lucero, M. J., Chesñevar, C. I., & Simari, G. R. (2009). On the accrual of arguments in defeasible logic programming. In *Proceedings of IJCAI* 2009 (pp. 804–809).
- 38. Grove, A. (1988). Two modellings for theory change. *The Journal of Philosophical Logic*, 17, 157–170.
- Hansson, S. O. (1993). Reversing the levi identity. The Journal of Philosophical Logic, 22, 637–669.
- 40. Hansson, S. O. (1994). Kernel contraction. The Journal of Symbolic Logic, 59, 845–859.
- 41. Hansson, S. O. (1999). A textbook of belief dynanics: Theory change and database updating. Kluwer Academic Publishers.
- 42. Kern-Isberner, G. (1999). Postulates for conditional belief revision. In *Proceedings of 16th International Joint Conference on Artificial Intelligence, IJCAI 1999* (pp. 186–191). Morgan Kaufmann.
- 43. Kern-Isberner, G. (2004). A thorough axiomatization of a principle of conditional preservation in belief revision. *Annals of Mathematics and Artificial Intelligence*, 40(1–2), 127–164.
- 44. Kern-Isberner, G. (2008). Linking iterated belief change operations to nonmonotonic reasoning. In G. Brewka, & J. Lang (Eds.), *Proceedings of 11th International Conference on Knowledge Representation and Reasoning, KR 2008* (pp. 166–176). Menlo Park: AAAI Press.
- 45. Levi, I. (1977). Subjunctives, dispositions, and chances. Synthese, 34, 423–455.
- 46. Lindström, S. & Rabinowicz, W. (1989). Epistemic entrenchment with incomparabilities and relational belief revision. In A. Fuhrmann, & M. Morreau (eds.), *The logic of theory change. Lecture Notes in Computer Science* (Vol. 465, pp. 93–126). Springer.
- 47. Makinson, D., & Gärdenfors, P. (1991). Relations between the logic of theory change and nonmonotonic logic. *Lecture Notes in Computer Science*, 465, 183–205.
- 48. Moguillansky, M. O., Rotstein, N. D., Falappa, M. A., García, A. J., & Simari, G. R. (2008). Argument theory change applied to defeasible logic programming. In *Proceedings of the 23rd Conference on Artificial Intelligence, AAAI 2008* (pp. 132–137).
- 49. Moguillansky, M. O., Rotstein, N. D., Falappa, M. A., García, A. J., & Simari, G. R. (2011). Argument theory change applied to defeasible logic programming. *Theory and Practice of Logic Programming (TPLP)*, (in press).
- Paglieri, F., & Castelfranchi, C. (2006). The toulmin test: Framing argumentation within belief revision theories (pp. 359–377). Berlin: Springer.
- 51. Peppas, P. (2004). The limit assumption and multiple revision. *Journal of Logic and Computation*, 14(3), 355–371.
- 52. Pollock, J. L. (1994). Justification and defeat. Artificial Intelligence, 67, 377–407.
- 53. Pollock, J. L., & Gillies, A. S. (2000). Belief revision and epistemology. *Synthese*, 122(1–2), 69–92.
- Prakken, H. (1993). Logical tools for modelling legal argument. Ph.D. thesis, Vrije Universiteit, Amsterdan.
- 55. Rahwan, I., & Simari, G. R. (2009). Argumentation in artificial intelligence. Springer.
- Rotstein, N. D., Moguillansky, M. O., Falappa, M. A., García, A. J., & Simari, G. R. (2008). Argument theory change: Revision upon warrant. In *Proceedings of the Computational Models of Argument, COMMA* '2008 (pp. 336–347).



- 57. Rott, H. (1992). Preferential belief change using generalized epistemic entrenchment. *The Journal of Logic, Language and Information*, 1, 45–78.
- 58. Simari, G. R., & Loui, R. P. (1992). A mathematical treatment of defeasible reasoning and its implementation. *Artificial Intelligence*, 53, 125–157.
- Tamminga, A. (2004). Expansion and contraction of finite states. Studia Logica, 76(3), 427–442.
- 60. Verheij, B. (1995). Accrual of arguments in defeasible argumentation. In *Proceedings of the 2nd Dutch/German Workshop on Nonmonotonic Reasoning*, (pp. 217–224). Utrecht.
- 61. Williams, M.-A. (1994). Transmutations of knowledge systems. In J. Doyle, E. Sandewall, & P. Torasso (Eds.), *Proceedings of 4th International Conference on Principles of Knowledge Representation and Reasoning (KR 1994)*, (pp. 619–629). Bonn, Germany.
- 62. Williams, M.-A. (1996). Towards a practical approach to belief revision: Reason-based change. In L. C. Aiello, J. Doyle, & S. C. Shapiro (Eds.), Proceedings of 5th International Conference on Principles of Knowledge Representation and Reasoning (KR 1996), (pp. 412–420). Cambridge, MA.

